Author's response to reviews

Title: Impairment of intellectual functions after surgery and posterior fossa irradiation in children with ependymoma is related to age and neurologic complications

Authors:

Katja von Hoff (hoff_k@klinik.uni-wuerzburg.de)
Virginie Kieffer (kieffer@igr.fr)
Jean-Louis Habrand (habrand@igr.fr)
Chantal Kalifa (kalifa@igr.fr)
Georges Dellatolas (dellatolas@vjf.inserm.fr)
Jacques Grill (grill@igr.fr)

Version: 5 Date: 21 January 2008

Author's response to reviews:

Dear Editor,

The requested changes have been made.

References list has been completed of all names of the authors and moved at the right place.

Tables have been reformatted as requested.

Tables and figures are now cited in the text as follows:

Table 1 on page 5, last paragraph
Table 2 on page 10, second paragraph
Figure 1 on page 10, last paragraph
Table 3 on page 11, risk factor analysis
Figure 2 on page 11, last paragraph
Figure 3 on page 12, second paragraph
Figure 4 on page 13, first line.

We hope we have done this satisfactorily for you,

Kind regards,

Jacques GRILL